

ORCA - Online Research @ Cardiff

This is an Open Access document downloaded from ORCA, Cardiff University's institutional repository:https://orca.cardiff.ac.uk/id/eprint/142321/

This is the author's version of a work that was submitted to / accepted for publication.

Citation for final published version:

Mithan, H. T., Hales, T. C. and Cleall, P. J. 2021. Topographic and ground-ice controls on shallow landsliding in thawing Arctic permafrost. Geophysical Research Letters 48 (13), e2020GL092264. 10.1029/2020GL092264

Publishers page: http://dx.doi.org/10.1029/2020GL092264

Please note:

Changes made as a result of publishing processes such as copy-editing, formatting and page numbers may not be reflected in this version. For the definitive version of this publication, please refer to the published source. You are advised to consult the publisher's version if you wish to cite this paper.

This version is being made available in accordance with publisher policies. See http://orca.cf.ac.uk/policies.html for usage policies. Copyright and moral rights for publications made available in ORCA are retained by the copyright holders.





Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL092264

Key Points:

- Arctic shallow landslides initiate in convergent topography immediately above the drainage network
- Melting of concentrated ground-ice at the base of the active layer is a primary control on Arctic shallow landsliding
- Deviations from the average thaw depth push thaw fronts into zones of concentrated ground ice in permafrost causing shallow landsliding

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

H. T. Mithan, huwmithan@gmail.com

Citation:

Mithan, H. T., Hales, T. C., & Cleall, P. J. (2021). Topographic and ground-ice controls on shallow landsliding in thawing Arctic permafrost. *Geophysical Research Letters*, *48*, e2020GL092264. https://doi.org/10.1029/2020GL092264

Received 23 DEC 2020 Accepted 1 JUN 2021

© 2021. American Geophysical Union. All Rights Reserved.

Topographic and Ground-Ice Controls on Shallow Landsliding in Thawing Arctic Permafrost

H. T. Mithan¹, T. C. Hales², and P. J. Cleall³

¹Department of Geosciences, National Taiwan University, Taipei, Taiwan, ²School of Earth and Environmental Sciences, Cardiff University, Cardiff, UK, ³School of Engineering, Cardiff University, Cardiff, UK

Abstract An increase in Arctic shallow landsliding is a potential consequence of climate warming. Warmer summer-air temperatures and larger rainfall events drive heat into the active layer, melting ice and decreasing soil shear stress. Topography has the potential to exacerbate landsliding by controlling the distribution of ground ice and the movement of water in the subsurface. We demonstrate that shallow Arctic landslides initiate in zero-order drainage basins consistent with models of shallow landsliding in non-permafrost environments. However, the low average slopes and low concavity of Arctic hillslopes cannot create pore-water pressures high enough to generate landsliding. Instead, two-dimensional slope stability modeling suggests that the vertical distribution of ground-ice distributions controls landslide susceptibility. High ground-ice concentrations close to the potential failure plane act as a stronger control than high average ice volumes or rapid thawing. Our results demonstrate that landslide susceptibility is strongly affected by topographic controls on ground ice and hydrology.

Plain Language Summary With a rapidly warming climate, landslide activity is increasing in the Arctic. Warmer Arctic temperatures thaw the permanently frozen soil (permafrost) that underlies hillslopes. Thawing creates meltwater, decreasing the strength of the soil, and in some cases initiating landsliding. We mapped the location of shallow landslides in Alaska. These landslides demonstrate a clear spatial pattern, forming above the channel network in depressions called zero-order drainage basins. The similarity between the spatial patterns of permafrost landslides and rainfall-triggered landslides suggests that the topography, hydrology, and spatial distribution of ground ice act to enhance landslide potential in zero order drainage basins. Modeling of hydrology and ground ice effects on slope stability, suggests a weak hydrologic control and a strong ground ice, in particular its concentration at potential landslide failure planes. We conclude that the strong relationship between topography and landsliding in the Arctic could improve estimates of shallow landslide susceptibility.

1. Introduction

Exceptional weather events are becoming more frequent in the Arctic causing permafrost to thaw rapidly (Dobricic et al., 2020; Lewkowicz & Way, 2019). Extreme summer air-temperatures and more frequent precipitation events drive thaw fronts deeper into ice-rich permafrost soils (Stocker et al., 2013), melting previously thermally stable ice. Melting ice in thawing permafrost increases pore-water pressures within the active layer that destabilize the overlying soil causing retrogressive thaw slumps (Balser et al., 2014; Lewkowicz & Way, 2019), solifluction (Harris et al., 2011), and active-layer detachment (ALD) failures (Lewkowicz, 1990). When these mass-wasting processes occur in proximity to populated areas they damage infrastructure (Hanna et al., 1998), cause economic loss (Allard et al., 2012), and threaten public safety (Larsson, 1982). By 2050, two-thirds of populated areas in permafrost regions will be affected by mass-wasting processes associated with permafrost degradation (Hjort et al., 2018). Long-term studies of permafrost degradation show that mass-wasting processes facilitate and enhance the mobilization of stored carbon stored in permafrost soils (Lamoureux & Lafreniere, 2009). Permafrost contains twice the amount of carbon than is currently in the atmosphere (Hugelius et al., 2014), making landsliding a potential contributor to the growing greenhouse gas emissions from degrading permafrost (Lamoureux & Lafrenière, 2018). A better understanding of the processes controlling hillslope instabilities will help quantify the contribution of mass-wasting processes to Arctic greenhouse gas emissions and evaluate their threat to infrastructure and public safety.

Assessing the susceptibility of Arctic hillslopes to mass-wasting processes is particularly challenging due to the strong control of ground ice on pore-water pressures in the active layer (Harris et al., 2011). Of the mass-wasting processes, ALDs are particularly susceptible. ALDs are shallow translational landslides that fail along a shear-plane between the base of the active layer and the top of the permafrost (Lewkowicz & Harris, 2005). Prior to ALD failure, annual plug-like solifluction movements of active-layer soil (Harris & Lewkowicz, 2000; Harris et al., 2008; Lewkowicz & Harris, 2005) progressively reduces basal shear strengths to residual values (Harris et al., 2008; Lewkowicz & Harris, 2005). When prolonged periods of high summer air-temperatures drive thaw fronts beyond the average active-layer depth and into the ice-rich layers above the permafrost melting ice lenses and collapsing pore spaces elevate excess pore-water pressures (Harris & Lewkowicz, 2000; Harris et al., 2011). The excess pore-water saturates the already weakened active-layer soil reducing shear-strengths further, creating a shearing plane and failure (Harris et al., 2001; Harris & Lewkowicz, 2000). The high excess pore-water pressures and residual shear strengths mean that ALDs can happen in low-relief topography, and on slopes as shallow as 3° (Rudy et al., 2016; Swanson, 2012). The importance of topography on ALD generation is typically discussed in the context of aspect controls on active layer thickness. North-facing slopes have a prevalence for higher rates of ALD failures due to thinner active layers that increase the likelihood of ground being ice closer to the surface, making them more susceptible to thaw (Rudy et al., 2016). Additionally, north-facing slopes have a greater amount of snow accumulation that remains on the slope later into the thawing season. This leads to a source of moisture for ground-ice aggradation during autumn freeze-back (Rudy et al., 2016; Young et al., 1997).

In non-permafrost settings, topography controls shallow landsliding through both the relationship between slope and shear stress and the hillslope hydrology, which controls pore-water pressure (Montgomery & Dietrich, 1994). Here, convergent topography drives throughflow that increases pore-water pressures and triggers shallow landslides because the infiltration of intense precipitation (Iverson, 1997). Given the complex interaction between ground-ice aggradation/degradation and topography, we still have a developing understanding of their relationship to landsliding in thawing permafrost (Harris & Lewkowicz, 1993; Mackay, 1981). In this study, we seek to understand the spatial distribution of ALD failures and whether this can provide insight into their triggering mechanisms. We perform a topographic landslide-analysis based on a mapped ALD inventory, comparing this to a model of slope stability under different ground-ice conditions to infer a mechanism for shallow landsliding on permafrost hillslopes.

2. Methods

2.1. Topographically Based Model

We developed an ALD inventory for a 100 km² region in the central Brooks Range (68.01°N, 155.85°W) (Figure 1a). It is a region of continuous cold-permafrost with low-relief topography that is overlain by till-derived colluvium and covered in Arctic tundra vegetation (Balser, 2015; Jorgenson, 2009; Trochim, Jorgenson et al., 2016). Permafrost extends to a depth of 295 m (GTN-P, 2021) and the active layer varies in thickness between 57 and 200 cm (Balser, 2015; CALM, 2021).Widespread ALD failures occurred in this region after a period of exceptionally high summer-air temperatures during the summer of 2004 (Swanson, 2010, 2014). We mapped 188 ALD failures (Figure 1a) using Maxar Technologies 2008-derived imagery accessed via Google Earth with a spatial resolution of 50 cm. Each ALD has a minimum mapped unit of 25 m². We defined an ALD as a soil-exposed runout zone with a compressional tongue of soil material at its foot (Figure 1b).

We compared our mapped ALD polygons to topographic derivatives of a 5-m interferometric synthetic-aperture radar-derived digital elevation model (available at USGS, 2015). To establish the importance of convergent topography to ALD failures we created a series of stream networks with upstream drainage area thresholds ranging from 12,500 to 125,500 m² at intervals of 2,500 m². Mapped ALD rasters were compared to these stream networks, which we buffered to account for the width of each ALD. Typically, stream networks with ranges of 25,000–50,000 m² mapped most closely to the streams visible on the imagery. Buffer widths varied from 10 to 100 m in increments of 10 m for each stream network and represent the hillslope distance above a given stream network. We assessed the spatial coincidence of ALDs and the stream network by comparing the area of ALD failures and non-failed terrain (Figure 1c).





Figure 1. (a) Localities of mapped active-layer detachments (ALDs) on top of a flow accumulation map (derived from the DEM using the D8 method) and areas with factors of safety ≤ 1 . (b) Sample of ALD failures in our study site from Google Earth imagery. Contour values are in meters above sea level. (c) Contour plot showing the area the landscape that is captured by varying sizes of stream networks (colored). The white contours represent the area of ALD failures captured within the varying sizes of stream networks.

We compared the distribution of ALD failures with a commonly used topographically derived landslide susceptibility model called SHALSTAB (Dietrich & Montgomery, 1998; Montgomery & Dietrich, 1994). SHALSTAB calculates landslide susceptibility using the infinite-slope assumption for soil-mantled land-scapes (where a thin soil overlay bedrock). In the model, the degree of topographic convergence dictates the pore-water pressure generated by throughflow (Montgomery & Dietrich, 1994). The model was applied by assuming that the permafrost represented an impermeable surface (analogous to a bedrock surface in a non-permafrost landscape), allowing us to assess the importance of throughflow in the generation of ALDs. We implemented the model using peak and residual soil shear-strength parameters (Table A1 and Figure S1) derived from geotechnical observations of ALD failures of Harris and Lewkowicz (2000) and hydraulic conductivity from Anderson and Anderson (2010).

2.2. Slope-Stability Model

We explored hypotheses for how topographic controls on ground ice might promote ALD failures by modeling slope stability. The hypotheses we tested were: (a) higher volumes of ice lenses along the permafrost-active layer boundary drive failure, (b) vertical distribution of ice-lens in the active-layer where ice lenses are concentrated at the base of the active layer are more likely to fail, and (c) excess pore-water dissipation rates caused by advancing thaw fronts strongly control failure potential. To design our hillslope and implement our slope-stability tests we used the geotechnical software GeoStudio (Krahn, 2004). For our stability analysis we applied the 2-D limit equilibrium-based Morgenstern-Price method of slices (Morgenstern & Price, 1965) which outputs a factor of safety (FoS) value to indicate the level of stability. The value derived from a FoS calculation is, in simple terms, the ratio between the maximum resisting forces and the destabilizing forces along a selected shear surface within a hillslope. The condition where the FoS = 1 means that the shearing resistance is fully mobilized and in equilibrium with the destabilizing forces, a FoS > 1 implies a stable slope, whilst a FoS < 1 implies failure. Where the FoS > 1 the slope is considered closer to failure the closer the value is to 1. As there is no field-based data to parameterize the model, we used the dimensions and soil geotechnical properties from well-constrained laboratory tests of ALD failures (Harris et al., 2008). By modeling a laboratory slope we can test our hypotheses regarding the role of ice lens distributions on ALD initiation while avoiding uncertainties in the material parameters and failure dimensions of a natural slope.

Our modeled hillslope was 15×7 m with a gradient of 24° and a maximum active-layer thickness of 1.5 m (normal to the surface). The active layer consisted of 18,060 cells with each cell being 5×2.5 cm. We simulated the passage of thawing from the surface to a depth of 1.5 m at increments normal to the hillslope surface. Between 0 and 1 m, thaw depths were considered at increments of 12.5 cm and below 1 m at increments of 5 cm (Figures 2a and S2). For each analysis, the thaw-front depth defined an interface between an upper layer of thawed soil and a lower layer of impermeable permafrost. We randomized, at any particular depth, the location of cells that represent ice lenses. When thawed, cells with no ice lenses were given a unit weight of 19.62 kN m⁻³, cohesion of 3.5 kPa, an internal angle of friction of 31° and geostatic pore-water pressures (Harris et al., 2008). Melting ice lenses generate pore-water pressures 20% greater than the hydrostatic (Harris et al., 2008; Kern-Luetschg & Harris, 2008), so as the thaw front passed each depth increment all of the cells identified as ice lenses were given elevated excess pore-water pressure values. Apart from our third test, excess pore-water pressures remained elevated throughout the test. In order to exclusively understand the role of ice lenses in ALD failures we omitted plug-like flow in our hillslope model by assigning peak values instead of residual values for our basal shear strengths.

To test our first hypothesis, we controlled the volume of ice lenses at a depth of 1.5 m in three experiments. First, we controlled the location of ice-lens cells, considering an even distribution (Figure 3a, blue line). Second, we randomly varied excess pore-water pressures by 0%–20% of the hydrostatic value in each ice lens cell (Figure 3a, orange line). Third, for each test we calculated the mean of five tests with a random distribution of ice-lens width (which is a function of the occurrence of collocated ice-lens cells) (Figure 3a, black line and gray range).

For the second hypothesis, we modeled the depth distribution of ice lenses based on three distributions of ground ice; uniform, linear, and power law (Figures 2b–2d). We varied the total volume of ice within the soil column, first considering scenarios with the same total volume of ice within the soil column (2,230 ice-lens cells) and second by varying the total volume of ice lenses to maintain a 60% ice volume at a depth of 1.5 m. Excess ice volumes greater than 60% of the soil mass are rare in the field (Allard et al., 1996; Kokelj & Burn, 2003; Morse et al., 2009), so we chose this as a maximum value for all tests. For our uniform distribution we randomly assigned 2,230 ice-lens cells evenly through the active layer to reach an average of 12% by volume (Figure 2b). For our linear distribution we allocated 2,230 ice-lens cells linearly with depth reaching 25% at 1.5 m (Figure 2c). Linear and uniform distribution of ice lenses represent end-member scenarios for ground ice distributions rather than any particular formation from two-sided freezing, these distributions help to put into context the results from our more representative power-law distribution. To account for concentrated ice lenses at the base of the active layer in permafrost regions with two-sided freezing, we fitted a power-law function of the form $I_{vol} = 11.841^{23.995}$ to field data from Kokelj and Burn (2003),





Figure 2. (a) Hillslope model dimensions for the power-law distribution of ice lenses during the factor of safety tests. Other models are in Figure S2. (b–d) Concentration of ice lenses through the active layer for our hillslope stability tests.

(Figure 2d), where z is a negative depth below the ground surface (m) and I_{vol} is the volume of ice relative to the soil mass (%).

For the third hypothesis we considered three tests designed to represent a zero, slow, and fast rate of excess pore-water dissipation (the rate of drainage of water from the thaw front) (Figure S4). We represented the dissipation of elevated excess pore-water pressure by returning pore-water pressures to hydrostatic values once outside of the excess pore-water pressure layer. In test 1 we assumed that the dissipation of excess pore-water pressure was zero (rapid thaw advance). For test 2 excess pore-water pressures were limited to cells within 12.5 cm of the thaw front (a slower thaw advance). In test 3, the excess pore-water pressure layer thickness was decreased to 2.5 cm (the slowest thaw advance). To contextualize the excess pore-water pressure layer, we used data from Harris et al. (2008, Figure 19b). From this we calculated the average dissipation time of 43.97 h for excess pore-water pressures and combined this with the reported average thaw rate of 1.48 mm hr⁻¹. We calculate that their experiment had an excess pore-water pressure layer thickness of 6 cm.





Figure 3. (a) The impact of varying ice lens concentration on the factor of safety along a failure plane at 1.5 m depth. The gray shaded area is the standard deviation of the five tests. (b) Slope stability results for a fast thaw advance where 2,230 ice lenses are concentrated in the active layer based on the three distribution scenarios.

3. Results and Discussion

ALD failures form within convergent topography in the uppermost parts of catchments (drainage areas of less than 75,000 m²; Figures 1a, 1c, and S3). Figure 1c shows a correlation between the location of ALD failures and the channel network relative to hillslopes (represented by the different buffer widths). Here, 40% of the ALD failures in the landscape occur within 15 m of the channel network at drainage areas greater than 50,000 m², this increases to a maximum of 76% at a drainage area of 12,500 m², while only occupying 22% of the total land area (Figures 1c and S3). Increasing the buffer width also increases the area of representative ALD failures but requires a buffer that covers considerably greater land area. A 14 m buffer is representative of the average ALD failure widths (Figure 1c) in our study area (14.2 \pm 3.8 m) and within the range of previously mapped ALD failures (10–30 m) in the Arctic (Swanson, 2014). Drainage densities in our study site are low, consistent with other permafrost catchments, where stream networks are ephemeral (Arp et al., 2012). While Arctic landscapes do not show the typical ridge and valley topography of non-permafrost landscapes (Grieve et al., 2018; Hack & Goodlett, 1960), the coincidence of landsliding and drainage area (Figure 1c) suggests that catchment topography influences landsliding.

In non-permafrost landscapes, the relationship between catchment topography and landsliding is usually associated with the hillslope hydrological controls on throughflow (Hack & Goodlett, 1960; Montgomery & Dietrich, 1994). We explored the potential for a similar hydrologic control using a landslide susceptibility map derived from SHALSTAB. The resulting map performed poorly because the slope angles upon which ALDs fail are much lower than shallow landslides in non-permafrost landscapes (Figures 1a and S1). SHALSTAB and similar models are driven by a steady state sub-surface hydrology, where throughflow saturates the soils and generates hydrostatic pore-water pressures. In our permafrost example, pore-water pressures created by the model are not sufficient to initiate failure at lower slope angles despite studies showing throughflow in this region in the thawed active layer (Evans et al., 2020; Kane et al., 2013; Voytek et al., 2016). We conclude, based on this experiment, that a simple hydrological mechanism cannot fully explain the propensity of ALDs to initiate in areas of convergent topography. With studies indicating that regions of convergent topography have higher ground ice content (Balser, 2015; Trochim, Prakash et al., 2016) we developed a series of tests to better understand how melting ground ice can trigger ALD failure. To do this we controlled the distribution and rate of ice lens melting in a 2-D slope-stability model.

We tested the role of ground ice in generating excess pore-water pressures that can lead to lower FoS values by modifying the distribution and rate of ice-lens melt in a 2-D slope-stability model. Our experiment shows that the distribution of excess pore-water pressures and the size of ice lenses along a failure plane strongly influence the FoS (Figure 3a). We observe that increasing the proportion of ice lenses along the failure plane causes a linear reduction in the FoS, with the positioning of ice lenses on the plane having a small impact on the FoS (Figure 3a). For our test with a constant excess pore-water pressure of 17.6 kPa, the minimum FoS was 1.005, while variable excess pore-water pressures yielded a higher minimum FoS of 1.065. The random distribution of ice-lens width affects the FoS most when the proportions of ice lenses along the failure plane are between 20% and 80%. At low and high proportions, the variability in ice lens size is small, as all ice lenses are either very small (single pixels) or very large. For intermediate proportions of ice lenses, lower factors of safety are associated with more frequent wide ice lenses, suggesting that having wide lenses with larger gaps between them produces lower factors of safety than evenly spaced, narrow ice lenses. This implies that slow and prolonged freezing of this zone (which is liable to produce thicker, more continuous lenses) may precondition slopes to failure by increasing ice content and therefore raising pore-water pressure well above the hydrostatic. Of the three variables tested (width, ice lens size distribution, and magnitude of excess pore-water pressure), the modeled slope is most sensitive to the magnitude of excess pore-water pressure generated by each lens. For example, with an average excess pore-water pressure value of 16.1 kPa, our random scenario generates a minimum FoS of 1.063 (Figure 3a orange line), compared with a FoS of 1.005 for a consistent excess pore-water pressure of 17.6 kPa (Figure 3a, blue line). ALD failures are therefore sensitive to the amount of ice along the failure plane and the excess pore-water pressures that it generates.

When we account for the depth distribution of ice lenses into our model, we found that the magnitude of excess pore-water pressures generated from melting ice lenses at depth controls the FoS. Linear and uniform ice lens distributions generate the most stable slopes with factors of safety 15%–20% higher than for a power distribution (Figure 3b). The difference in the FoS reflects the proportion of ice at the failure plane; with low ice-volume in the uniform (12%) and linear (25%) distributions relative to the power-law distribution (60%) (Figure 3b). This result is confirmed when we set the ice-lens concentration to 60% at the failure plane for every distribution scenario (Figure S2). At these high ice-concentrations, the uniform and linear distributions produce lower factors of safety because they have higher ice-concentrations through the active layer. Therefore, it is not the total volume of ice lenses in the active layer that governs stability, instead it is the proportion of ice lenses found along the failure plane. This numerical experiment provides constraint on the potential ground ice conditions that would be necessary to trigger ALDs.

With most observed ALDs initiating shortly after extreme warming events (Lewkowicz & Harris, 2005), we examined if the rate at which a thaw front advances into the active layer will impact slope stability. Surprisingly, the rate made no measurable difference to the FoS in our model runs (Figure S4), as the peak excess pore-water pressure is more important than the rate of pore-water pressure dissipation. Hence the number and size of ice lenses on that failure plane has a stronger effect on the failure potential. Our results demonstrate a strong spatial control on the initiation of ALDs, that could provide a framework for the development of landslide susceptibility maps in the Arctic. ALDs form within a narrow range of slopes and drainage areas within our study area. These areas tend to be convergent, and a few hundred meters upslope of the channel heads. Hence, it may be possible to model landslide susceptibility as a simple function of those parameters (as demonstrated in Figure 1). A greater understanding of the spatial and topographic controls on ground-ice formation is necessary to improve the physical basis for potential landslide susceptibility modeling. In particular, the variation of ground ice concentrations as a function of drainage area in a catchment may be a key variable in future estimates of landslide susceptibility.

While there is no explicit temporal component to our modeling, the observation that ALD failure depends on the development of large ice lenses at depth, has implications for the temporal distribution of landslides (Figure 4). If we consider the mass of ice at a potential failure plane as the balance between ice generated at a particular depth by cryosuction and the total amount of thaw within the summer, then landslide susceptibility will vary, even within a steady state climate due to random weather events. We can explore the temporal evolution of landslide potential in a changing climate (Figure 4). At average thaw depths there is a low concentration of ice lenses because in any given year there is an equal likelihood of melting or expanding ice lenses. In a steady state climate, a warmer summer, where the active layer is deeper (possibly two or





Figure 4. (a) A conceptual model showing the impact of rising summer-air temperatures on the magnitude of active-layer detachment failures and (b) its relationship with thaw depth and ice lens concentration at the base of the active layer.

three standard deviations above the average) than the average, will increase the probability of a landsliding event. When average summer air-temperatures rise due to a warming climate, average thaw fronts deepen, meaning that a smaller deviation from the average is required to generate an ALD failure event. In the tenth year of our conceptual model (Figure 4), extended high summer air-temperatures push the maximum thaw depth beyond the base of the active layer and into a region of concentrated ground ice in the upper permafrost. Our conceptual model can help explain the observed concurrent nature of ALD failure events after periods of higher than average warm air temperatures observed in low-relief Arctic landscapes (Lamoureux & Lafrenière, 2018; Lewkowicz & Harris, 2005; Swanson, 2014). Our observations show that ALD failures have a propensity to initiate in zero-order drainage basins (convergent topography) and that this failure is likely related to a topographic control on ground-ice concentrations. This suggests that landslide susceptibility models for ALD failures can be developed based on simple topographic metrics (slope and drainage area), but that going beyond susceptibility to understand hazard requires an understanding of the temporal (and spatial) development of ground ice.

4. Conclusion

Our observations show that ALD failures have a propensity to initiate in zero-order drainage basins and that landslide triggering is likely related to a topographic control on ground-ice concentrations in the active layer at the permafrost table. We found that a non-linear distribution of ice lenses in the active layer produces a less stable slope at the failure plain. Conceptually, this has implications for the temporal evolution of shallow landslide potential. As average thaw fronts deepen, only a small deviation from the average thaw depth is required to push thaw fronts into zones of concentrated ice lenses in the upper permafrost. This suggests that landslide susceptibility models for ALD failures can be developed based on simple topographic metrics, but that going beyond susceptibility to understand hazard requires an understanding of the temporal and spatial development of ground ice in Arctic hillslopes.



Data Availability Statement

Datasets for this research are available in this in-text data citation references: Mithan et al. (2020), with a CC BY 4.0 license.

References

- Allard, M., Caron, S., & Begin, Y. (1996). Climatic and ecological controls on ice segregation and thermokarst: The case history of a permafrost plateau in Northern Quebec. *Permafrost and Periglacial Processes*, 7(3), 207–227. https://doi.org/10.1002/ (sici)1099-1530(199609)7:3<207::aid-ppp219>3.0.co;2-4
- Allard, M., Lemay, M., Barrette, C., L'Hérault, E., Sarrazin, D., Bell, T., & Doré, G. (2012). Permafrost and climate change in Nunavik and Nunatsiavut: Importance for municipal and transportation infrastructures. In Nunavik and Nunatsiavut: From science to policy. An Integrated Regional Impact Study (IRIS) of climate change and modernization (pp. 171–197).
- Anderson, R., & Anderson, S. (2010). Geomorphology: The mechanics and chemistry of landscapes. Cambridge: Cambridge University Press. Arp, C. D., Whitman, M. S., Jones, B. M., Kemnitz, R., Grosse, G., & Urban, F. E. (2012). Drainage network structure and hydrologic behavior of three lake-rich watersheds on the Arctic Coastal Plain, Alaska. Arctic, Antarctic, and Alpine Research, 44(4), 385–398. https://
- doi.org/10.1657/1938-4246-44.4.385
 Balser, A. W. (2015). Retrogressive thaw slumps and active layer detachment slides in the Brooks range and foothills of Northern Alaska: Terrain and timing. University of Alaska Fairbanks. https://doi.org/10.13140/RG.2.1.4497.6160
- Balser, A. W., Jones, J. B., & Gens, R. (2014). Timing of retrogressive thaw slump initiation in the Noatak Basin, northwest Alaska, USA. Journal of Geophysical Research: Earth Surface, 119(5), 1106–1120. https://doi.org/10.1002/2013jf002889
- CALM (2021). Circumpolar Active Layer Monitoring (CALM) site U26. Retrieved from https://www2.gwu.edu/~calm/data/north.htm
 - Dietrich, W. E., & Montgomery, D. R. (1998). SHALSTAB: A digital terrain model for mapping shallow landslide potential. University of California.
 - Dobricic, S., Russo, S., Pozzoli, L., Wilson, J., & Vignati, E. (2020). Increasing occurrence of heat waves in the terrestrial Arctic. Environmental Research Letters, 15(2), 024022. https://doi.org/10.1088/1748-9326/ab6398
 - Evans, S. G., Godsey, S. E., Rushlow, C. R., & Voss, C. (2020). Water tracks enhance water flow above permafrost in upland Arctic Alaska hillslopes. Journal of Geophysical Research: Earth Surface, 125(2), e2019JF005256. https://doi.org/10.1029/2019JF005256
 - Grieve, S. W. D., Hales, T. C., Parker, R. N., Mudd, S. M., & Clubb, F. J. (2018). Controls on zero-order basin morphology. Journal of Geophysical Research: Earth Surface, 123(12), 3269–3291. https://doi.org/10.1029/2017JF004453

GTN, -P. (2021). Global Terrestrial Network for Permafrost (GTN-P) site US 12. Retrieved from http://gtnpdatabase.org/boreholes/view/1007/ Hack, J. T. T., & Goodlett, J. C. (1960). Geomorphology and forest ecology of a mountain region in the central Appalachians. USGS Profession-

- al Paper (Vol. 347). Retrieved from https://pubs.er.usgs.gov/publication/pp347%0Ainternal-pdf://228.60.152.3/report-2.pdf Hanna, A. J., McNeill, D., Tchekhovski, A., Fridel, T., & Babkirk, C. (1998). The effects of the 1994 and 1995 forest fires on the slopes of the
- Norman Wells pipeline. In Proceedings 7th international permafrost conference (pp. 421–426). Yellowknife, NWT. Harris, C., Kern-Luetschg, M., Christiansen, H. H., & Smith, F. (2011). The role of interannual climate variability in controlling solifluction processes, Endalen, Svalbard. Permafrost and Periglacial Processes, 22(3), 239–253. https://doi.org/10.1002/ppp.727
- Harris, C., & Lewkowicz, A. G. (1993). Form and internal structure of active-layer detachment slides, Fosheim Peninsula, Ellesmere Island, Northwest-Territories, Canada. Canadian Journal of Earth Sciences, 30(8), 1708–1714. https://doi.org/10.1139/e93-149
- Harris, C., & Lewkowicz, A. G. (2000). An analysis of the stability of thawing slopes, Ellesmere Island, Nunavut, Canada. Canadian Geotechnical Journal, 37(2), 449–462. https://doi.org/10.1139/cgj-37-2-44910.1139/t99-118
- Harris, C., Rea, B., & Davies, M. (2001). Scaled physical modelling of mass movement processes on thawing slopes. *Permafrost and Periglacial Processes*, 12(1), 125–135. https://doi.org/10.1002/ppp37310.1002/ppp.373

Harris, C., Smith, J. S., Davies, M. C. R., & Rea, B. (2008). An investigation of periglacial slope stability in relation to soil properties based on physical modelling in the geotechnical centrifuge. *Geomorphology*, 93(3–4), 437–459. https://doi.org/10.1016/j.geomorph.2007.03.009

- Harris, C. R., Millman, K. J., vander Walt, S. J., Gommers, R., Virtanen, P., Cournapeau, D., et al. (2020). Array programming with NumPy. Nature, 585(7825), 357–362. https://doi.org/10.1038/s41586-020-2649-2
- Hjort, J., Karjalainen, O., Aalto, J., Westermann, S., Romanovsky, V. E., Nelson, F. E., et al. (2018). Degrading permafrost puts Arctic infrastructure at risk by mid-century. *Nature Communications*, 9(1), 5147. https://doi.org/10.1038/s41467-018-07557-4
- Hugelius, G., Strauss, J., Zubrzycki, S., Harden, J. W., Schuur, E. A. G., Ping, C. L., et al. (2014). Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps. *Biogeosciences*, 11(23), 6573–6593. https://doi.org/10.5194/ bg-11-6573-2014
- Hunter, J. D. (2007). Matplotlib: A 2D graphics environment. Computing in Science & Engineering, 9(3), 90–95. https://doi.org/10.1109/ MCSE.2007.55
- Iverson, R. M. (1997). The physics of debris flows. Reviews of Geophysics, 35(3), 245-296. https://doi.org/10.1029/97RG00426
- Jorgenson, M. T. (2009). An ecological land survey and landcover map of the Arctic Network. US Department of the Interior, National Park Service, Natural Resource Program Center.
- Kane, D. L., Yoshikawa, K., & McNamara, J. P. (2013). Regional groundwater flow in an area mapped as continuous permafrost, NE Alaska (USA). *Hydrogeology Journal*, 21(1), 41–52. https://doi.org/10.1007/s10040-012-0937-0
- Kern-Luetschg, M., & Harris, C. (2008). Centrifuge modelling of solifluction processes: Displacement profiles associated with one-sided and two-sided active layer freezing. *Permafrost and Periglacial Processes*, 19(4), 379–392. https://doi.org/10.1002/ppp.633

Kokelj, S. V., & Burn, C. R. (2003). Ground ice and soluble cations in near-surface permafrost, Inuvik, Northwest Territories, Canada. Permafrost and Periglacial Processes, 14(3), 275–289. https://doi.org/10.1002/ppp.458

- Krahn, J. (2004). Stability modeling with SLOPE/W: An engineering methodology. Calgary, Alberta, Canada: GEOSLOPE/W International Ltd.
- Lamoureux, S., & Lafreniere, M. (2009). Fluvial impact of extensive active layer detachments, Cape Bounty, Melville Island, Canada. Arctic, Antarctic, and Alpine Research, 41(1), 59–68. https://doi.org/10.1657/1938-4246(08-030)[lamoureux]2.0.co;2
- Lamoureux, S. F., & Lafrenière, M. J. (2018). More than just snowmelt: Integrated watershed science for changing climate and permafrost at the Cape Bounty Arctic Watershed Observatory. *Wiley Interdisciplinary Reviews: Water*, 5(1), e1255. https://doi.org/10.1002/wat2.1255

Acknowledgments

We would like to thank two anonymous reviewers and Professor Michael Gooseff for their helpful comments which improved our manuscript. Thank you to the python community for programming support that enabled us to analyze and plot our data (Harris et al., 2020; Hunter, 2007; Met Office, 2015; Neteler et al., 2012). This work is supported by Cardiff University President's Scholarship and the Ministry of Science and Technology in Taiwan under the grants 109-2116-M-002 -028 and 109-2811-M-002 -676.

- Larsson, S. (1982). Geomorphological effects on the slopes of Longyear valley, Spitsbergen, after a heavy rainstorm in July 1972. *Geografiska Annaler: Series A, Physical Geography*, 64(3–4), 105–125. https://doi.org/10.2307/520639
- Lewkowicz, A. G. (1990). Morphology, frequency and magnitude of active-layer detachment slides, Fosheim Peninsula, Ellesmere Island, NWT. Proceedings of the 5th canadian permafrost conference (Vol. 54, pp. 111–118).
- Lewkowicz, A. G., & Harris, C. (2005). Frequency and magnitude of active-layer detachment failures in discontinuous and continuous permafrost, northern Canada. *Permafrost and Periglacial Processes*, *16*(1), 115–130. https://doi.org/10.1002/ppp.522
- Lewkowicz, A. G., & Way, R. G. (2019). Extremes of summer climate trigger thousands of thermokarst landslides in a High Arctic environment. *Nature Communications*, 10(1), 1329. https://doi.org/10.1038/s41467-019-09314-7
- Mackay, J. R. (1981). Active layer slope movement in a continuous permafrost environment, Garry Island, Northwest-Territories, Canada. Canadian Journal of Earth Sciences, 18(11), 1666–1680. https://doi.org/10.1139/e81-154

Met Office (2015). Cartopy: A cartographic python library with a Matplotlib interface. Retrieved from http://scitools.org.uk/cartopy

- Mithan, H. T., Hales, T. C., & Cleall, P. J. (2020). Data input and results for modelling the topographic and ground-ice controls on shallow landsliding in thaving Arctic permafrostitle. https://doi.org/10.6084/m9.figshare.13475838
- Montgomery, D. R., & Dietrich, W. E. (1994). A physically-based model for the topographic control on shallow landsliding. Water Resources Research, 30(4), 1153–1171. https://doi.org/10.1029/93wr02979
- Morgenstern, N. R., & Price, V. E. (1965). The analysis of the stability of general slip surfaces. *Géotechnique*, 15(1), 79–93. https://doi. org/10.1680/geot.1965.15.1.79
- Morse, P. D., Burn, C. R., & Kokelj, S. V. (2009). Near-surface ground-ice distribution, Kendall Island Bird Sanctuary, Western Arctic Coast, Canada. Permafrost and Periglacial Processes, 20(2), 155–171. https://doi.org/10.1002/ppp.650
- Neteler, M., Bowman, M. H., Landa, M., & Metz, M. (2012). GRASS GIS: A multi-purpose Open Source GIS. Environmental Modelling & Software, 31, 124–130. https://doi.org/10.1016/j.envsoft.2011.11.014
- Rudy, A. C. A., Lamoureux, S. F., Treitz, P., & van Ewijk, K. Y. (2016). Transferability of regional permafrost disturbance susceptibility modelling using generalized linear and generalized additive models. *Geomorphology*, 264, 95–108. https://doi.org/10.1016/j. geomorph.2016.04.011
- Stocker, T. F., Qin, D., Plattner, G. K., Tignor, M., Allen, S. K., Boschung, J., et al. (2013). *IPCC, 2013: Summary for policymakers in climate change 2013: The physical science basis, contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change.*
- Swanson, D. (2010). Mapping of erosion features related to thaw of permafrost in Bering Land Bridge National Preserve, Cape Krusenstern National Monument, and Kobuk Valley National Park.
- Swanson, D. (2012). Mapping erosion features to thaw of permafrost in the Noatak National Preserve. Alaska.
- Swanson, D. K. (2014). Mapping of erosion features related to thaw of permafrost in the NPS Arctic Inventory and Monitoring Network, Alaska. *Natural Resource Data Series NPS/ARCN/NRDS-2010*, 122.
- Trochim, E. D., Jorgenson, M. T., Prakash, A., & Kane, D. L. (2016). Geomorphic and biophysical factors affecting water tracks in Northern Alaska. *Earth and Space Science*, *3*(3), 123–141. https://doi.org/10.1002/2015ea000111
- Trochim, E. D., Prakash, A., Kane, D. L., & Romanovsky, V. E. (2016). Remote sensing of water tracks. *Earth and Space Science*, 3(3), 106–122. https://doi.org/10.1002/2015ea000112
- USGS (2015). U.S. Geological Survey, 20150804, USGS NED Original Product Resolution AK IFSAR-D6-L2-C60 2013 ArcGrid 2015. Retrieved from https://www.sciencebase.gov/catalog/item/581d2302e4b08da350d55d18
- Voytek, E. B., Rushlow, C. R., Godsey, S. E., & Singha, K. (2016). Identifying hydrologic flowpaths on arctic hillslopes using electrical resistivity and self potential. *Geophysics*, 81(1), WA225–WA232. https://doi.org/10.1190/GEO2015-0172.1
- Young, K. L., Woo, M. K., & Edlund, S. A. (1997). Influence of local topography, soils, and vegetation on microclimate and hydrology at a High Arctic site, Ellesmere Island, Canada. Arctic and Alpine Research, 29(3), 270–284. https://doi.org/10.2307/1552141